Author’s response to reviews

Title: Rapidly progressive Neovascular glaucoma following coronary artery bypass graft surgery in a patient with type 1 diabetes mellitus

Authors:

Vasant Raman (vasant.raman@nhs.net)

Rosemary Lambley (rosemary.lambley@doctors.org.uk)

Version: 1 Date: 03 Jan 2018

Author’s response to reviews:

1. At the end on the introduction the reviewer has asked to add a paragraph.

   This has been added.

2. Past medical, and other relevant history has been added, I am not sure about his environmental history.

3. The patient was not admitted to the Eye department, but was treated as an outpatients. In view of this he had did not have CBC, liver function or any other test as he was being treated purely for his eye condition only. The internal medicine were looking after his general condition.

4. As suggested by the reviewer, the discussion starts with a summarizing statement.